Author's response to reviews

**Title:** Nurse-based case management for aged patients with myocardial infarction - study protocol of a randomized controlled

**Authors:**

Inge Kirchberger (Inge.Kirchberger@helmholtz-muenchen.de)
Christa Meisinger (Christa.Meisinger@helmholtz-muenchen.de)
Hildegard Seidl (Hildegard.Seidl@helmholtz-muenchen.de)
Rupert Wende (Rupert.Wende@klinikum-augsburg.de)
Bernhard Kuch (Bernhard.Kuch@klinikum-augsburg.de)
Rolf Holle (holle@helmholtz-muenchen.de)

**Version:** 2  **Date:** 17 May 2010

**Author's response to reviews:** see over
BMC Geriatrics

Augsburg, 17.05.10

Corresponding person Telephone Fax E-mail
Dr. Inge Kirchberger +49-821-400-4176 +49-821-400-2838 Inge.Kirchberger@helmholtz-muenchen.de

Ref: Revision of the manuscript “Nurse-based case management for aged patients with myocardial infarction: study protocol of a randomized controlled trial”

Dear editorial team,

We have performed the following changes according to your request:

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

➢ Please note that the affiliations are already in the requested format. Maybe it is confusing, but “Institute of Epidemiology” and “Institute of Health Economics …” are the departments of the Helmholtz Zentrum München, which is the institution.

References: please cite all references in numerical order in the text. Several citations do not appear in the correct position.

➢ We have modified the order of the references accordingly.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, “The Title of My Prestigious Publication: A Scientific Study.” should be changed to “The title of my prestigious publication: a scientific study.”
We have changed the title to “Nurse-based case management for aged patients with myocardial infarction: study protocol of a randomized controlled trial”

Title page: remove wordcount.

- Sorry, but we could not retrieve a wordcount on the title page.

Corresponding author: one or more corresponding author(s) must be indicated on the title page, and their email address must be given.

- The email address of the corresponding author was added.

Abstract: please add the trial registration number to the abstract in the submission system.

- The trial registration number was included in the abstract in the submission system.

Underlining: please remove.

- Underlining was deleted.

References: please add the heading References.

- The heading was added.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

- The figure was modified.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

- Typography was checked.

Looking forward to hearing from you.

Yours sincerely,

Inge Kirchberger, PhD, MPH